

Is Kropotkin's Mutual Aid the Antidote to the Predatory Journal Problem?

DU ROCHER, Andrew

Available from Sheffield Hallam University Research Archive (SHURA) at:

https://shura.shu.ac.uk/35170/

This document is the Published Version [VoR]

Citation:

DU ROCHER, Andrew (2025). Is Kropotkin's Mutual Aid the Antidote to the Predatory Journal Problem? tripleC: Communication, Capitalism & Critique., 23 (1), 47-69. [Article]

Copyright and re-use policy

See http://shura.shu.ac.uk/information.html



Is Kropotkin's Mutual Aid the Antidote to the Predatory Journal Problem?

Andrew R. du Rocher

Sheffield Hallam University, Sheffield, UK, <u>andy.r.du.rocher@gmail.com</u>, <u>a.du-rocher@shu.ac.uk</u>

Abstract: Academic capitalism has encouraged the development of pay-to-publish journals, and may have also encouraged the proliferation of poor-quality profit motivated predatory journals. Predatory journals undermine the confidence that people have in scientific research, and have created an ethical crisis. Alternatives to capitalist ideologies can reveal how an anti-capitalist intervention to the predatory journal problem might be developed. A response to the predatory journal problem might be developed using a collaborative behaviour referred to as mutual aid. Mutual aid is an organisational component of the anarchist communism proposed by Russian dissident, geographer, zoologist, and anarchist Pyotr Alexeyevich Kropotkin. Networks of non-commercial not-for-profit online open access publishing houses and journals could be developed by faculty in higher education institutions using a mutual aid strategy. It is entirely possible that a gradual and sustained increase in (anti-capitalist) online open access journals.

Keywords: capitalism, predatory journals, academic publishing, platform socialism, anarchist communism, mutual aid, intervention, open access, Kropotkin

1. Introduction

Higher education institutions and their academics are involved in the production of, and use of, research articles that appear in academic journals (Niles et al. 2020). Publishers of research journals have proliferated. There has been an increase in what are generally considered honest quality scholarly publishers, but there has also been an increase in dishonest poor-quality journals. These dishonest poor-quality profitmotivated journals have been labelled "predatory journals". Often predatory journals charge article processing fees that are higher than what is required for the purpose of publication, without offering quality peer review or editorial services in return. This indicates that these organisations function like businesses that are not particularly motivated by academic scholarship (Kurt 2018). Reasons for academics publishing in these journals include fear of academic job loss and the pressure to publish, career competition in the higher education environment, and a lack of awareness that the journals are dishonest or of poor quality (Demir 2018, Kurt 2018).

The activities of predatory publishing organisations corrupt the ideas behind legitimate open access publishing (Ferris and Winker 2017). Many predatory journals offer open access publication in return for financial payment, therefore the link with open access can cause a bias against quality open access journals (Krawczyk and Kulczycki 2021). Poor quality articles from predatory journals can also be naively captured in the systematic review process, which could be detrimental to research impact (Rice, Skidmore and Cobey 2021). Predatory journals undermine the confidence that the readership and the public have in research (Ferris and Winker 2017). It is thus no surprise that the predatory journal problem has been described as an ethical crisis (Cress and Sarwer 2019).

It has been suggested that interventions to this problem could be based on the creation of quality monitoring and/or journal ranking systems (Dadkhah and Bianciardi 2016; Wallace 2019), and/or the organisation of international editing collaborations (Callaghan and Nicholson 2020). It has also been suggested that more academic publishing could be conducted by learned societies and/or university presses (Teixeira da Silva et al. 2022).

The question this paper asks is: Is Kropotkin's mutual aid the antidote to the predatory journal problem? The article presents how an intervention to the predatory journal problem might evolve that is based upon non-commercial open access scholar-led publishing, and how such an intervention might be organised by academics in higher education institutions. Specifically, section 2 discusses how traditional academic publishing is located within a system of market capitalism. Section 3 analyses some socialist alternatives to capitalist ideologies. Section 4 introduces the anarchist-communist thought of Russian dissident, geographer, zoologist, and anarchist Pyotr Alexeyevich Kropotkin (1842-1921). Section 5 discusses who might organise a response to the predatory journal problem. Section 6 suggests what an anticapitalist intervention to the predatory journal problem might look like. Section 7 provides some important monetary considerations. The overall focus of this paper is to consider how an intervention to the predatory journal problem might be developed using a collaborative behaviour referred to as mutual aid. Mutual aid is a component of the anarchist communism proposed by Kropotkin (1902, 1906).

2. Academic Publishing and Capitalism

Society can be considered as the totality of social systems that are external to the individual. The social horizon is the social background that lies behind the everyday perception of each individual person. From this perspective, segments of this social horizon can implicitly influence, mould, and shape a person's behaviour. Thus, a capitalist society can implicitly influence perception and a person's behaviour (Gunderson 2021). Capitalism pushes social process and institutions to create an organised system of the production of goods as a profitable commodity. Thus, capitalism manifests a society in its own image (Marx and Engels 1969/1848). It is a capitalist imperative to organise the rapid production of these profitable commodities in a large volume (Stoner 2020).

Although academic publishing started off as a profession, it became a business within a system of market capitalism, and finally an industry organised using capitalist logic (Luescher and van Schalkwyk 2018). It has been suggested that there is a tendency for academic capitalism to develop within economies that are already dominated by capitalist markets (Jessop 2018). Jessop (2018) suggests that academic capitalism tends to develop in five analytically separable stages that also have the potential to overlap. Stage 1 is the commercialisation stage where research and education are produced as commodities and are offered for sale:

"This stage involves various forms of simple commodity production that are not yet subject to the competitive pressures of profit-oriented, market-mediated capitalist production to reduce the socially necessary labor time for production and the turnover time of capital" (Jessop 2018, 105).

Stage 2 is the capitalisation stage where the capitalist market economy develops in academia, which reinforces the commercialisation process. In this stage Jessop (2018) suggests that one can observe:

"free trade in knowledge, the rationalization of its production based on tight control on costs and their recovery, and universities and research institutes using their own accumulated capital and/or loans to boost revenues" (Jessop 2018, 105).

Stage 3 is described by Jessop (2018) as a further step towards capitalisation that is based on a more radical modification of the process of production. In this stage the means of production is separated from the contribution of intellectual material:

"It involves the quasi-commodification of mental labor as an input, including the separation of intellectual labor from the means of intellectual production. This contributes to hierarchization and precarization of intellectual labor, loss of professional status, and attempts to limit the freedom of teaching and research in the interests of cost reduction and profit maximization" (Jessop 2018, 105).

Stage 3 also includes

"the appropriation of traditional knowledge, privatization of the intellectual commons, commodification of teaching materials, scholarship, scientific research, and scientific publications, and, more recently, digitization of lectures enabling their virtually costless reproduction and circulation – while charging consumers for access" (Jessop 2018, 105).

Stage 4 is the financialisation stage in which Jessop (2018) describes two components:

"First, mobile capital enters these fields considered as sites of profitable substantive investment; and, second, market forces equalize profits across specific universities, colleges, and research enterprises and between these and other fields of potentially profitable investment" (Jessop 2018, 106).

Stage 5 is described by Jessop (2018) as a system that is dominated by financial interests. In this stage,

"a finance-dominated system subordinates education and research to the profitability requirements of capital as property" (Jessop 2018, 106).

Throughout the stages of academic capitalism described by Jessop (2018), the products that are produced for sale include learning and teaching materials, actual tuition itself, and research. It is the production of research that is of relevance to the present discussion. It is easy to see how the observation of the development of academic capitalism would attract mainstream journal publishers to the market. Critically, this may have also encouraged the proliferation of predatory journals. The stages of academic capitalism described by Jessop (2018) offer a clear entrance into the market for profit motivated organisations. In short, within academia there is a system already in place whereby there is a separation of intellectual input from the means of intellectual production, where intellectual content is a commodity offered for sale, and where capitalism is the accepted norm.

It has been suggested that academic research is now driven by the need for academics to continually publish, as opposed to academic research driving the need to publish (Hyland 2023). Capitalist open access research communication organisations simply see the content of academic production as goods to be commercially produced, distributed, and consumed (Knoche 2020).

Considering that academic publishing has become an industry organised by capitalist logic (Luescher and van Schalkwyk 2018), it seems likely that this has encouraged the proliferation of predatory journals and their host publishing organisations. It therefore makes intuitive sense to consider some political ideologies that are opposed to capitalism that can inspire an intervention to the predatory journal problem.

3. An Alternative to Capitalist Ideologies

Capitalism appeared at some point in the era spanning the fourteenth and sixteenth centuries when human behaviour shifted from using markets sporadically, to producing goods for them as an all-consuming behaviour (Sunkara 2020). Capitalism involves people indirectly and anonymously exchanging products which is achieved by using money. In this situation, members of society do not mutually contribute to each other's social activities, they just loot each other to the best of their abilities. By contrast, a Marxist inspired socialist perspective suggests that the recognition of the needs of societal members should out way individual members personal egotism. This can be achieved if products are not exchanged via a market using money (Honneth 2018).

Throughout the middle of the twentieth century, socialism and capitalism were contesting for the opportunity to define the future (Sunkara 2020). Historically, socialist ideas have revolved around the idea that society should control economic activity. Socialist economic models were based on cooperation, community, and association, and were predicated on the ideas that individual self-fulfilment is dependent upon the self-fulfilment of others (Honneth 2018). However, socialism in the twentieth century can be considered a false start, as confidence in socialism was gradually destroyed during the 1980s. Marxist perspectives were confined to classrooms in universities. Marx was considered a philosopher as opposed to a revolutionary, and capitalism was considered to have won as the final form of humanities socio-cultural evolution (Sunkara 2020).

Capitalism is resilient but is susceptible to crisis. The inequalities created by capitalism are provocative enough to bring about forms of resistance. However, many people lack any reason to think that politics can make their life any better. Moreover, collective activity in or outside the workplace carries a greater risk to those involved than an acceptance of the status quo (Sunkara 2020). This still leaves open the following question:

"But what about the end goal of socialism – extending democracy radically into our communities and workplaces, ending the exploitation of humans by other humans?" (Sunkara 2020, 237).

The current quandary for socialism is to understand how to direct discontent at the inequalities created by capitalism and mould the discontent into a viable confrontation to the capitalist system. Training new socialist organisers is important as new generations of writers, speakers, and thinkers are needed, as any form of socialism is unachievable without them (Sunkara 2020). Socialism has been described as a form of historical experimentalism (Honneth 2018). 19th century socialists spent most of their time writing about possible recipes for a different future. By contrast, contemporary socialists need to convince people that the future can be different,

without knowing what that future may look like. Today socialism has been revived, but it still needs to construct a coherent narrative of opposition to economic elites and to challenge capitalism (Sunkara 2020). If socialism is to survive it needs to be reactivated in a more contemporary format. Social freedom may be achieved via experimenting with different methods of creating economic value beyond that which is defined by capitalism, and these new methods should involve cooperative mechanisms (Honneth 2018). In this context, Honneth says:

"The guideline for any experimentation with different economic combinations must lie in strengthening 'the social' in the economic sphere as much as possible, enabling all those involved to satisfy their needs through complementary activity without compulsion or restricted influence" (Honneth 2018, 67).

The current organisation of the market must be dissected into its differential components so that they can be examined for suitability for cooperative manifestations of economic organisation. Thought experiments resulting from this can inform the creation of a revised form of socialism and the expansion of economic social freedom. Historical experimentalism suggests that the more preliminary methods that are tested under real-world economic situations the more influence they will have in the context of future political and practical activities (Honneth 2018). These socialist experiments could include experimental ways of combating the predatory journal problem, with the experiments being conducted by researchers and academics.

In a true socialist society, the means of production should be in common ownership of the producers (Marx 1844). The advent of digital technologies has facilitated the creation of a new form of commodity, and related means of production, that has been exploited by capitalist markets. A form of digital capitalism has developed whereby digital communication, and knowledge is seen as a commodity, and where the means of communication (the means of production) are seen as private property (Boucas 2020, Fuchs 2020a). This has, however, encouraged the manifestation of a form of digital socialism (Fuchs 2020a). The difference between capitalist and socialist ideas on communication technologies can be summed in the following way:

[In digital capitalism,] knowledge and communication are privately controlled and owned by the few as private property, whereas in a socialist society knowledge and communication technologies are gifts and common goods that are collectively produced and owned" (Fuchs 2020a, 20).

A socialist critique of digital capitalism needs to address the digital alienation caused by the lack of common control of digital resources and commodities, and the lack of common control of the creation of digital ideologies on the internet (Fuchs 2020a). To summarise, in digital capitalism (or communicative capitalism) computer hardware and software, as well as apps and data are the means of production. When these are owned by capitalist monopolies economic power is centralised. Digital socialism aims to break up such monopolies. Digital socialism asks that technology is used to benefit everybody and calls for the creation of forms of non-commercial collective and public digital communication services (Fuchs 2020a). Critically, the current system of digital capitalism does not allow members of society to increase their engagement with socially beneficial work, or to increase autonomous peer production, therefore these are the goals that digital socialism needs to address (Boucas 2020). It might be possible gain some inspiration "from literary communist utopias for the creation and organisation of communicative and digital socialist society and a utopian Internet" (Fuchs 2020b, 146). In summary, one way of combating digital capitalism / platform capitalism would be for the digital means of production to be owned by members of society, which would result in a form of digital socialism (Fuchs 2021).

Many schisms occur within the broader socialist ideology, but one ideology of relevance here is termed socialist anarchism or social anarchism. Social anarchism involves solidarity and voluntary collaboration, and should not be confused with individualistic types of anarchism (Kinna 2019). The media and the state often distort perceptions of anarchism and associate it with chaos. However, social anarchism can be considered a rationalised constructive reaction to hierarchy and domination (Ferretti 2016). A key component of social anarchism (but not individualistic or capitalist anarchism, and therefore not anarchism in general) is its opposition to capitalism, opposition to authoritarianism, and its focus on the decentralisation of organisational decision making. These anarchists seek to replace the capitalist strategies of exploitation and profit-seeking, with mutual aid strategies (Williams 2018).

The social anarchist strategy of prefiguration involves blending anarchist ideologies into practical endeavours which can result in the organisation of alternatives to mainstream social institutions, such as counter-institutions which function as cooperatives (Cornell 2016). Prefigurative political behaviour is the future-orientated creation of alternative political situations and/or the reflection of political goals in current activities. Thus, prefiguration involves the creation of alternative practices in the anticipation of a changed situation in the future (Yates 2021). Such political strategies aim to establish radical new practices that challenge the status quo (Törnberg 2021).

Some people might be described as implicit anarchists and may identify with general anarchistic labels such as anti-authoritarianism or libertarian socialism. By contrast, other people may form anarchist collective groups and explicitly identify with specific ideological perspectives such as anarcho-communism or anarcho-syndicalism (Williams 2018). Discussion now turns to the anarchist communism advocated by Pyotr Alexeyevich Kropotkin.

4. Pyotr Alexeyevich Kropotkin's Thought

4.1. The Conquest of Bread

The Conquest of Bread (Kropotkin 1906) was first published in French as La conquête du pain in 1892. In the Conquest of Bread, Kropotkin (1906) describes so-called "civilised societies" as a minority of monopolists taking advantage of their alleged rights that have been historically obtained to take two-thirds of the products produced by society and squander them. This minority prevents the remainder of society from producing the products that are required by all, as they are obligated to produce the products that provide the largest profit for the monopolist minority. According to Kropotkin, "in this is the substance of all socialism" (Kropotkin 1906, 5). Kropotkin considered that the result of this system is that enterprise and the resulting production increases the profit of the speculator but does not consider the requirements of the community.

Kropotkin (1906) notes that the privileges of the minority of monopolists and speculators are upheld by a system of law enforcement, which manifests a system fraught with lies and corruption. He continues by discussing how huge amounts of labour are wasted in producing and funding an unnecessary lavish and depraved lifestyle for the rich and fashionable, which is funded by consumers who are pushed into purchasing what they do not require. Therefore, in such a system, huge amounts of labour are routinely wasted in producing wares that are profitable but injurious. Moreover, this squandering activity occurs at the detriment of the production of useful products that would benefit the other two-thirds of society. Similarly, in contemporary society, academic capitalism appears to have encouraged the creation of predatory journals who can publish substandard research publications. When predatory journals publish substandard research, it can be naively used as the basis of future genuine attempts at research. The future research that is based upon the substandard research will then have been conducted and possibly published, at the detriment of the production of useful research (based on quality prior research) that could benefit the rest of society.

Kropotkin (1906) says that a key problem of the capitalist system is that production produces surplus-value in the first place. Thus, the issue with capitalism is not solely that any surplus-value goes to the capitalists who own the means of production. Similarly, in contemporary society, academic capitalism appears to encourage the creation of predatory journals that publish large amounts of (potentially surplus) research in order to maximise profits. Kropotkin says that society needs to aim to produce as many goods as are required to ensure the well-being of all its members, whilst also minimising the waste of energy of its members. Kropotkin summarised this issue in the following manner:

"Might it not be that production, having lost sight of the *needs* of man, has strayed in an absolute wrong direction, and that its organization is at fault? And as we can prove that such is the case, let us see how to reorganize production so as to really satisfy all needs" (Kropotkin 1906, 79).

In contemporary society, the re-organisation of production in order to satisfy societal needs could be interpreted as the re-organisation of academic publishing to increase the availability of open access journals that do not charge any fees to contributing authors or the journal readership. The anarchist communism described by Kropotkin can be seen as a description of a post-scarcity society that would be highly productive and thus improve the means of production. Therefore, in contemporary society the anarchist communism described by Kropotkin can be reinterpreted to incorporate the proposed utilisation of digital technologies as a means of increasing production (Fuchs 2020b). The utilisation of digital technologies can be used to increase the availability of open access journals that do not charge any fees to authors or readers, as well as increasing useful production in general.

In *The Conquest of Bread*, Kropotkin (1906) says that the privilege and riches of the minority must be redistributed as common property to serve the collective interests of all in society. Kropotkin argues that all members of society have a right to live, and that society ought to share with all its members whatever is required for existence. Kropotkin says that the wage system will perish when the capitalist system of commercial production perishes. He writes that novel organisations should be created in a multitude of forms that are all based upon the communist principle of "to every man according to his needs" (Kropotkin 1906, 15). He argues that communistic behaviour intermittently appears in societies already. He discusses examples concerning the railways, such as the implementation of season tickets allowing unlimited travel, and zone systems whereby travel costs the same across long and short distances within the same zone. Kropotkin continues by explaining that a huge network of European railways evolved that consist of a multitude of different

companies, with multitudes of shareholders. These companies and their shareholders all interact and collaborate via reason and free agreement that is not enforced by government. Accordingly, Kropotkin asks "why, in the midst of our societies, consisting of groups of free workers, should we need a government?" (Kropotkin 1906, 57). Kropotkin's observations clearly suggest that collaborative and collective behaviour between groups of people throughout a large part of the globe is possible. This is what is required if a solution to the predatory journal problem is to be found. In *The Conquest of Bread*, Kropotkin (1906) envisages a society where all members have become producers, all members have an education, and thus all members have the time and the opportunity to develop the ability to create science or art. A key point made by Kropotkin (1906) is: "Society must itself, take possession of all means of production" (Kropotkin 1906, 42).

Kropotkin (1906) also aimed to eliminate the division between manual labour and mental labour. This division between manual labour and mental labour is clearly present in stage 3 of the development of academic capitalism as described above. To reiterate, in this stage, the means of production of intellectual material is separated from the contribution of intellectual material (Jessop 2018). Kropotkin's (1906) ideas on eliminating the division between manual labour and mental labour have also contributed to ideas on a new digital socialism. For example, a future based on digital socialism would allow a single person to produce digital media, use freely available information obtained via the Internet, participate in social activities (including online activities) and work on producing content that will be distributed online using creative commons licences, all on the same day (Fuchs 2020b). The idea of digital socialism is clearly applicable to organising a new publishing paradigm as an intervention to the predatory journal problem.

4.2. Mutual Aid: A Factor of Evolution

In *Mutual Aid: A Factor of Evolution*, Kropotkin (1902) focuses on the idea of collaboration. Kropotkin was inspired by his observations of the mutual aid-based cooperation in the animal kingdom. Concerning the observed advantage of mutual aid amongst animals he wrote:

"Don't compete! – competition is always injurious to the species, and you have plenty of resources to avoid it! That is the *tendency* of nature, not always realized in full, but always present. That is the watchword which comes to us from the bush, the forest, the river the ocean. 'Therefore combine – practise mutual aid! That is the surest means for giving to each and to all the greatest safety. The best guarantee of existence and progress, bodily, intellectual and moral" (Kropotkin 1902, 71).

Kropotkin (1902) was also inspired by the mutual aid-based co-operation in early human societies, cities, groups of poverty-stricken people, and labour movements. He was interested in the human history of mutual-aid institutions and considered how they still survive in society:

"It seems, therefore, hopeless to, look for mutual aid institutions and practises in modern society. What could remain of them? And yet, as soon as we try to ascertain how the millions of human beings live, and begin to study their everyday relations, we are struck with the immense part which the mutual-aid and mutualsupport principles play even now-a-days in human life. Although the destruction of mutual-aid institutions has been going on in practise and theory, for full three or four hundred years, hundreds of millions of men continue to live under such institutions; they piously maintain them and endeavour to reconstitute them where they have ceased to exist" (Kropotkin 1902, 193-194).

Kropotkin (1902) was inspired by the way human societies have always manifested practices and institutions that manage conflict and oblige the needs of their membership. Kropotkin argues that these practices and institutions did not arise from any centralised authority but were manifested by the customs and instincts of the people within society. Kropotkin's suggestion that mutual aid is based upon customs and instincts within society suggests that the customs and instincts of sections of society should manifest a solution to the predatory journal problem. Kropotkin was particularly interested in how the labouring classes utilise mutual aid type strategies to aid survival:

"For anyone who has any idea of the life of the labouring classes it is evident that without mutual aid being practised among them on a large scale they never could pull through all their difficulties" (Kropotkin 1902, 240).

In his work on mutual aid, Kropotkin (1902) discusses the idea that voluntary cooperation and mutual aid have more practical utility than egoistic behaviour. His ideas are still consistent with current biological theorising, as co-operation is observable wherever there is biological organisation. For example, co-operation is observable when studying genes and how they function within the genome, cells and how they form multi-cell organisms, and when studying humans and how they form societies (West et al., 2007). Kropotkin (1902) disagrees with the Darwinian concept of the "survival of the fittest" (Darwin 1859), that has often been used for justifying the existence of slavery, exploitation, and warfare. However, it is notable that Darwin (1859) accepted that the "struggle for existence" can also include one individual being dependent upon another (Pantel et al. 2023).

Kropotkin (1902) does not entirely reject the involvement of competition and individual selection for survival, but rather sees the importance of mutual aid and collaboration. Kropotkin considers collaboration to be significant factor in the struggle for existence. Kropotkin argues that competition between individuals in the same species is detrimental to any advantageous effect of cohabitation. For Kropotkin, mutual aid is a driving force of evolutionary change. His work is still considered relevant in the field of ecology and mutualism (Pantel et al. 2023). Indeed, as Pantel et al. discuss:

"From an evolutionary perspective, cooperation is a stable strategy when its benefits outweigh the benefits of cheating (where individuals in the group do not participate and still receive benefits)" (Pantel et al. 2023, 87).

The work of Kropotkin (1902) and Pantel et al. (2023) suggests that the advancement of human knowledge should be based on collaboration and co-operation. Evolutionary biology also suggests that co-operation evolves where the relationship is beneficial to all involved (although over time the mutual benefits may change). However, the system of co-operation needs to be resilient to the evolutionary effect of "cheaters" that benefit in some way without having co-operated. In the context of the current debate, when authors naively submit research to predatory journals, the predatory journals profit from the work of others without contributing very much. The magnitude of the predatory journal problem suggests that academic publishing has not yet become resilient to cheaters.

Species exist in an ecosystem that includes distinct components that manifest as different types of interactions that include predatory, parasitical, and competitive behaviour as well as co-operative, mutualist, and commensal behaviour (Pantel et al. 2023).

"No species can only participate in one of these interactions – all organisms consume, and organisms do share finite resources. However, organisms also produce resources for other species as byproducts, at no cost to themselves, and others exchange materials to cooperatively improve their fitness" (Pantel et al. 2023, 89).

The behaviour of predatory journals can be considered parasitic as they value profits over quality. However, researchers live in the same ecosystem and should be able to formulate a mutual aid-based solution to predatory journal problem. The works of Kropotkin suggest that mutual aid and sociality should be the components of a newly evolved social moral. Kropotkin envisages a future where people appreciate the repercussions of their behaviour and realise that they need to limit their own needs in order to not harm the rest of society. Kropotkin's ideas can therefore be considered a foundation for ethical behaviour (Padovan 1999) which might inform a response to the predatory journal problem. Afterall, the predatory journal problem has been described as an ethical crisis (Cress and Sarwer 2019). Kropotkin did not have any naive belief, or optimistic belief, that that human progress was inevitable, and he did not assume that evolutionary change was identical to progress. Kropotkin proposes that mutual aid is the key factor in the advancement of evolution, but he does not rule out the possibility that other factors such as the need for struggle, could lead to a reversal of evolutionary progress. Moreover, Kropotkin also discusses in a late nineteenth century newspaper article that the environment could alter the balance of mutual aid type behaviours and individualist behaviour (Kinna 1995). Nevertheless, re-examining the evolutionary ideas of Kropotkin, and building upon his ideas of mutual aid and co-operation, might elucidate some of the possibilities for social change that are needed to combat social crises (Pantel et al. 2023).

Mutual aid has become a fundamental element of left-wing political movements (Kenworthy et al. 2023). Mutual aid projects evolve when existing public services are not sufficient, are non-inclusive, or exacerbate state abuse (Spade 2020). Such behaviours are central to the survival of minority groups related to race, disability, sexuality, as well as groups of people affected by plague, pandemics, natural disasters, and illnesses such as HIV/AIDS (Kenworthy et al. 2023).

In summary, the literature on mutual aid suggests that subcomponents of anarchist communist ideology can be used to inform and guide small scale organised responses to societal problems. The line of thinking developed by Kropotkin in his work on mutual aid (Kropotkin 1902) can be interpreted as a theory of organisation that was way ahead of its time (de Geus 2014). Mutual aid projects often evolve to provide relief to people living through crises (Spade 2020), and the predatory journal problem has been described as an ethical crisis (Cress and Sarwer 2019). Thus, it is possible that mutual aid projects could evolve as an organised response to the ethical crisis of the predatory journal problem. There is already a requirement of mutual aid-based behaviours in the

academic publishing process. For example, peer review is not usually financially rewarded (Tötösy de Zepetnek and Jia 2014), and participation in unpaid peer review is a form of mutual aid (Springer et al. 2017). As Kropotkin states:

"it is especially in the domain of ethics that the dominating importance of the mutual-aid principle appears in full. That mutual aid is the real foundation of our ethical conceptions seems evident enough" (Kropotkin 1902, 249).

5. Who Can Organise a Response to the Predatory Journal Problem?

One way of tackling the predatory journal problem would be for more academic publishing to be organised by groups of academics, learned societies, or university presses (Teixeira da Silva et al. 2022). The suggestion by Teixeira da Silva et al. (2022) is based on the successful academic led publishing that is already common in countries such as Finland (Late et al. 2019; Pölönen et al. 2021) and South Korea (Hong and Youn 2020). There has also been a gradual increase in the development of new university presses (Lockett and Speicher 2016). It may be possible to liberate open access publishing from being predominantly controlled by industrialised capitalist publishers, by supporting a fully digitalised online open access publishing model that has an anti-capitalist ethos (Knoche 2020).

One publishing model that is appropriate here is the Diamond Open Access (DOA) model. The DOA model involves the online publication of material that is distributed without any monetary fees being charged to readers or authors (or libraries). The DOA model is run by non-commercial not-for-profit organisations, or networks of associations, and does not allow any for-profit or commercial re-use of published material. Although content published using the DOA model is typically in online digital format, DOA does allow publishers to charge printing costs (with no profit margin) if print copies are to be offered. The DOA model uses Creative Commons licenses which allows others to re-share (with the same terms) and to reuse content for noncommercial purposes. These creative commons licenses can either allow others to modify and expand on work or disallow such practices. The authors are always given credit under such licensing agreements. Some versions of creative commons licenses allow commercial reuse of material, but these versions can be excluded by the DOA model as they violate the not-for-profit ethos (Fuchs and Sandoval 2013). Despite the existence of DOA models, at present there does not appear to be enough not-for-profit and/or academic led online publishing houses and journals to function as a strong antidote to the predatory journal problem. Thus, there is still a need for the development of a more widespread non-commercial academic led open access publishing paradigm.

The workload of academics employed by higher education institutions involves teaching, research, publishing, and contributions to administration within university departments. However, some academics may also involve themselves in some form of academic activism, with the intent to facilitate some sort of social change (Kenny 2021). Higher education institutions can approve of the involvement staff in academic activism if the activity is not organised in opposition to the institution itself. If academic activism is aimed at facilitating a positive social change, then this activity can be presented by institutions as evidence of the relevance of universities and their staff (Kenny 2021).

It has already been discussed how faculty in higher education institutions can be involved in not-for-profit collaborative scholar-library-university led open access publishing (Adema and Moore 2018). This collaborative form of organisation can involve both horizontal and vertical processes. Vertical collaborations involve the interaction between university libraries, higher education institutions themselves, external funding bodies, and other organisations. By contrast, horizontal collaborations can involve consortia or collectives that are manifested via unions formed by networks of small independent presses and/or publishing communities that publish books or journals. This type of organisation can facilitate the provision of mutual aid and thus the sharing of practical support. Collectives involved in these networks can be non-hierarchical, and all members can be involved in decision making. Therefore, these organisations can be described as having a horizontal or flat organisational structure (Adema and Moore 2018).

The focus of the present discussion is to consider how an intervention to the predatory journal problem could be organised. The perspective on scholar-library-university led collaborative academic publishing discussed by Adema and Moore (2018) implicitly resonates with some of the thoughts on collaboration and mutual aid proposed by Kropotkin (1902, 1906). However, discussion now turns to how the anarchist communism proposed by Kropotkin (1902, 1906) can explicitly inform a mutual aid-based intervention to the predatory journal problem.

6. Anti-Capitalism and Open Access

Capitalism generates discontent which generates forms of resistance. Motivations for discontent with, and therefore resistance to capitalism can be related to class interests but can also be related to moral values. Thus, in the twenty-first century, anti-capitalism cannot be explained just in terms of Marxist social class struggle. Anti-capitalism can also be explained in terms of a struggle to defend one's moral values. Thus, in the twenty-first century, it is imperative to create anti-capitalist organisations based on moral values and not just social class interests (Wright 2021). Erik Olin Wright (2021) discusses five strategic components of historic anti-capitalist activity:

"smashing capitalism, dismantling capitalism, taming capitalism, resisting capitalism and escaping capitalism. Even though in practice these strategies intermingle, each of them constitutes a distinct way of responding to the harms of capitalism" (Wright 2021, 38).

Wright (2021) differentiates between two strategic dimensions of anti-capitalist activity. The first of these strategic dimensions includes the strategies of taming capitalism and resisting capitalism and forms a dimension that should result in neutralizing some of the harm caused by organised capitalism. The second dimension includes the strategies of smashing capitalism, dismantling capitalism, and escaping capitalism, and forms a dimension that should result in transcending the organisational structure of capitalism. A contemporary strategy for resisting capitalism appears to be evolving in parts of southern Europe and Latin America. This strategy of "eroding capitalism" merges bottom-up society-based strategies of resisting capitalism and escaping capitalism, with top-down state-based strategies of taming capitalism and dismantling capitalism. The strategy of eroding capitalism is predicated upon the thought that no economy is ever purely capitalist, as economic systems that are dominated by capitalism include pockets of alternative economic activity. This alternative economic activity includes activity where goods and services are produced and distributed using cooperative, collaborative, peer-to-peer, and/or not-for-profit behaviour. Thus, a capitalist economy can be seen as an ecosystem that includes a multitude of economic activities that are dominated by capitalism. The strategy of eroding capitalism therefore

involves expanding these non-capitalist economic activities into the gaps, niches, and cracks within the capitalist dominated economic ecosystem (Wright 2021).

Anti-capitalism can be considered an organisational form that changes depending upon circumstances. Political collectives can adopt a range of organisational forms that apply to anti-capitalism without explicitly using the label "anti-capitalism" to confirm their eagerness to adopt such organisational forms (Malherbe 2024). There are examples in society where cooperatives are run by workers who make decisions concerning finances and production. Consumer cooperatives exist where people purchase goods in bulk to cut cost, and then redistribute the goods amongst their members. Moreover, trading schemes exist where members receive credits that are earned and spent by providing and receiving services to and from other members (Gordon 2012). The existence of these behaviours in society suggests that faculty in higher education institutions should be capable of adopting similar strategies in any newly evolving publishing paradigm that emerges as a response to the predatory journal problem. Moreover, further inspiration and ideas can be acquired from the many internet platform cooperatives that already exist, as they can take different organisational forms (as listed and discussed in Scholz 2016). Of particular interest are those platform co-operatives that can be described as a produser-owned platform. Produser is a term which is used by Trebor Scholz to refer to a blend of user and producer owned platform. Platforms such as this are a response to the activities of monopolies such as Google or Facebook who attract users by offering a "free service" whilst profiting from their data and content. Produser platforms offer users opportunities for co-ownership of the site through which they are distributing their product (Scholz 2016). Such a system could be adapted to suit a not-for-profit mutual aid inspired academic publishing paradigm. Platform co-operatives rely on essential interaction with other co-operatives:

"Platform co-ops are not islands, entire of themselves. Every co-op is part of an eco system" (Scholz 2016, 21).

Platform co-operatives use a means of production that is in collective ownership. Therefore, they should be able to create a common collection of resources that members can draw from, and contribute to, according to their requirements and capabilities (Padimitropoulos and Malamidis 2024). Access to collections of shared resources would also aid the development of new journals in a newly developing mutual aid inspired not-for-profit online open access academic publishing paradigm. Indeed, platform co-operativism envisages a future where digital technology serves the social good. In such a future, users of a people's Internet have decision making and ownership rights, as well as the space to debate, communicate and create content (Sandoval 2020).

In his book *Envisioning Real Utopias*, Wright (2010) describes how real utopias concerning social empowerment and the economy can be organised and experimented with. Economic activity that is organised by the voluntary collaborative and/or collective behaviour of members of society can be described as a social economy (Wright 2010). Wright (2010) discusses Wikipedia as an example of a social economic behaviour, an example of anti-capitalist internet activity, and thus a real utopia. Nobody receives payment for entries in Wikipedia. Maintenance of its software is completed by volunteers. There is no charge for access to content in Wikipedia, and there are no advertisements. The key criticisms of Wikipedia concern the accuracy of the entries, the issue of neutrality within arguments and debates, and the problem of

intentional distortion of information (Wright 2010). In the context of the creation of new online open access journals, these problems should be resolved by mutual aid-based peer review. The question that needs considering here concerns whether it is possible to create a real utopia that reflects a not-for-profit online open access academic publishing paradigm.

One way of combating capitalism is to develop a co-operative market economy. Mutualist co-operatives can collaborate with each other via coordinated joint activity. However, co-operatives can be less efficient than capitalist businesses and can remain a small part of societies economy (Wright 2010). Nevertheless, the ideas discussed by Wright (2010) clearly resonate with those of Kropotkin. Wright draws attention to the following fact:

"The oldest vision for an emancipatory alternative to capitalism is the workerowned firm. Capitalism began by dispossessing workers of their means of production and then employing them as wage-laborers in capitalist firms. The most straightforward undoing of that dispossession is its reversal through workerowned firms" (Wright 2010, 165).

There are also some further principles of anarchist communism that are applicable here. Kropotkin (1906) implies here that all useful property and intellectual property should be collectively owned as individual products are the work of everyone:

"Science and industry, knowledge and application, discovery and practical realization leading to new discoveries, cunning of brain and hand, toil of mind and muscle -- all work together. Each discovery, each advance, each increase in the sum of human riches, owes its being to the physical and mental travail of the past and present. By what right then can any one whatsoever appropriate the least morsel of this immense whole and say – This is mine, not yours?" (Kropotkin 1906, 7).

Kropotkin (1906) clearly considers that the development of such property relies on the physical and intellectual labour of people who predated any individual's contribution (Kropotkin 1906). This is also clear when he says:

"Thousands of writers, of poets, of scholars, have laboured to increase knowledge, to dissipate error, and to create that atmosphere of scientific thought, without which the marvels of our century could never have appeared" (Kropotkin 1906, 6).

Similarly, newly produced academic work builds upon what is known beforehand. Thus, existing theories aid the development of new theories and are often the basis for new research. Moreover, one unitary journal article is not simply an individual piece of creativity, it is part of an ongoing communication process that is continued and contributed to by several people over many years (Fuchs and Sandoval 2013). This emphasises why all knowledge produced by research should be made freely available to everyone.

Kropotkin (1906) also argued that a need for a product should be felt prior to producing it as illustrated by his rhetorical question: "Is it not the study of needs that should govern production?" (Kropotkin 1906, 78). This is particularly relevant to the development of an intervention to the predatory journal problem. For example, only

research articles that are of good quality and beneficial to others need to be published. This can more easily be achieved via the organisation of a non-capitalist not-for-profit publishing paradigm, than by continuing with the current, and dominant, capitalist industrialised publishing paradigm.

If there is enough interest in an anti-capitalist mutual aid-based solution to the predatory journal problem, networks of new not-for-profit online publishing houses and open access journals could be developed by faculty in higher education institutions. Kropotkin's thoughts on mutual aid (Kropotkin 1902) and anarchist communism in general (Kropotkin 1906) can clearly inspire the development of such a small-scale publishing paradigm. When any new online publishing house or journal is developed, the means of production would be the sum of all resources that are used to produce the journals, which could be collaboratively owned (or at least collectively cared for, until being passed to the next generation of caretakers). The shared ownership of any means of production would mean that decisions concerning finances, production, and the method of interaction with other mutual aid networks, could be made collaboratively. A contemporary social anarchist economic perspective suggests that experimentation with different varieties of economic organisation can be tried, whilst maintaining the emphasis on production that is not solely motivated by profit (Wigger 2014). Such experiments would be congruent with the experimental socialism advocated by Honneth (2018), and aid the development of new not-for-profit online publishing houses and journals. Moreover, within contemporary models of social anarchism differing methods of production might co-exist, as might horizontally organised networks that are involved in the exchange of services and products (Wigger 2014). This is particularly relevant to the possible development of new not-for-profit online publishing houses and journals. Different publishing houses organised by different co-operatives, and located in different communities, or countries, might produce journals in a different way. However, they can still form horizontally organised networks that share services and products via mutual aid. The utility of horizontally organised mutual aid networks for the sharing of practical support relating to collaborative open access academic publishing has already been noted (Adema and Moore 2018).

As discussed above, social anarchists can work alongside non-anarchists to organise social movements that tackle societal problems and facilitate social change (Cornell 2016). Thus, any new online publishing houses adopting an anarchist(ic) approach may be able to collaborate with university presses and libraries located within higher education institutions. Such online publishing houses will be able to resist being acquired or taken over by university presses if they want to (Loacker 2021).

Furthermore, Adema and Moore (2018) discuss the formation of a radical open access collective, and the creation of the associated website. This is indicative of the increasing interest in collective and collaborative not-for-profit academic publishing. This type of activity is compatible with the prefiguration strategies used by social anarchists. As discussed above, these strategies involve blending social anarchist ideologies into practical activities which are intended to result in the future creation of alternative organisations, counter-institutions, and co-operatives (Cornell 2016).

In time, mutual aid networks could be developed nationally and internationally which could facilitate higher education faculty to develop new online publishing houses and journals. International collaborations might reduce international problems. For example, a problem exists whereby authors located in developing countries can feel that well known western journals are prejudiced against them, which can push them towards publishing in predatory journals (Kurt 2018). However, socialist experimental

interventions in one context should improve the likelihood of success of socialist experimental interventions in other contexts. If these interventions are successful, they can be tested worldwide. Thus, one can envisage the existence of a future collaborative network of successful socialist experimental interventions (Honneth 2018).

7. Monetary Considerations

Alternative political and economic activity can exist as pockets of organised low-profile resistance within the wider capitalist economy. Pockets of social and/or economic activity that are located outside of any capitalist processes and capitalist relations have been described as activity that is operating within an exilic space. In these exilic spaces novel practices and organisational structures can exist in autonomy from capitalist accruement and exist independently of social control (O'Hearn and Grubačić 2016). Thus, the staff of new online publishing houses and journals could interact within their organisation, or between similar organisations, using the economic practices of their choice. However, the organisational structures within these exilic spaces would need be able to interact with external organisational structures distributed across the wider capitalist economy.

An intervention to the predatory journal problem might be based on the organisation of new non-commercial online open access journals, that do not charge author fees. Despite the anticipated involvement of collaboration, mutual aid, and the social anarchist economies described above, there would still need to be the involvement of some money. Typically, journals incur costs of employing editorial, technical and design staff, but there are further costs of running electronic journals. These costs can include server and online infrastructure maintenance, as well as the employment of a troubleshooting team. However, electronic publishing can be organised for a lower financial cost than print publishing (Tötösy de Zepetnek and Jia 2014). Some of these maintenance costs might be able to be reduced if horizontally organised publishing networks are created that are involved in the exchange of the required services and products. For example, an online journal will require an article submission system and website to be maintained (Chiarelli et al. 2023). This could be achieved via mutual aid networks collaborating during setting up and running online publishing houses. Kropotkin (1902) also discusses the utility of mutual aid in creating human knowledge:

"the practise of mutual-aid and its successive developments have created the very conditions of society life in which man was enabled to develop his arts, knowledge, and intelligence; and that the periods when institutions based on the mutual-aid tendency took their greatest development were also the periods of the greatest progress in arts, industry, and science" (Kropotkin 1902, 247-248).

Kropotkin's ideas on mutual aid are clearly still relevant to the proposed development of online open access journals. There are, however, some monetary costs that may be less easily reduced by mutual aid strategies. The creation of, and maintenance of, an online open access journal will require funds to pay for web hosting, provision of a domain name, and article DOI registration. There are also other costs to consider such as membership fees for organisations that offer services relating to databases and metrics, as well as plagiarism detection software and support relating to publication ethics (Chiarelli et al. 2023). A means of funding the services that cannot be supplied via mutual aid networks and the social anarchist economies described above would need to be found.

Existing Internet platform co-operatives also have similar problems, and they also rely on sources of funding, designers and workers, solicitors, software programmers and software engineers. Internet platform co-operatives require a different type of funding option than mainstream enterprises, as conventional sources of funding are often unavailable to them. Co-operatives can receive funding from development funds and crowdfunding. However, disruptive regulatory bodies are often keen to stop the advancement of experiments into platform co-operativism. Platform co-operatives are reliant upon open-source software licences. Therefore, it has been proposed that more software needs to be made accessible for them to use. For example, software programmers could help this situation by developing and distributing core protocols which can be modified to suit different open-source applications with different end uses (Scholz 2016). Moreover, forms of Internet-based peer-to-peer collaborative production already exist within the wider capitalist economy and interact efficiently with capitalist organisations if required. Linux (an open-source operating system for computers) is used by capitalist organisations, and some of these organisations donate paid software engineer time to aid production of the software. This process occurs even though Linux has an open-source license that prevents any patenting of the software (Wright 2021). This type of help with software access would also help those developing new not-for-profit online open access journals and/or their host online notfor-profit online publishing houses, as a response to the predatory journal problem.

Publishers that are non-commercial, radical, and not part of mainstream publishing must survive financially within the capitalist market economy. Sometimes their requirement for economic survival can appear somewhat incongruent with their original political ambition. Collective publishing organisations can find that the cost of time and resources is high in comparison to their commercial counterparts. Thus, collective publishing organisations need to consider the amount of collective behaviour and participation that can realistically be provided at any one time, whilst existing in the capitalist market economy. Resolving goal conflict that arises between political ambition and economic requirements can be a dissonant experience (Comedia 1984).

A study on DOA journals (Bosman et al. 2021) has reported that the cost of running these journals varies substantially. A financial stability analysis indicated that 43% of the journals claimed that they break even, whereas 25% of them claimed that they made a financial loss. Moreover, 31% of them claimed that they do not know their financial status, and 1% make a profit. Of these journals 60% use volunteers, and 86% (of the 60% who use volunteers) have a medium to high reliance upon voluntary workers. Financial support and funding for these journals comes from a range of sources including (but not limited to) research organisations and universities, national funding agencies, publishers, learned societies. libraries. foundations/trusts. NGO/charities, and museums. Of these journals, 91% (1186) said that they did not charge monetary fees for any of the services they provide. Of those journals who did charge some monetary fees, printing fees (52), layout fees (43) or copy-editing fees (39) were the most common types of monetary fee. Respondents were asked how research funders could financially support DOA journals. 869 responses were received. Many respondents asked for support from institutions, funders, and funding platforms. Many asked for support for editorial staff, proof-reading staff, copy-editing staff, as well as for translation, and plagiarism detection software. Moreover, long term funding was asked for by 16% of the journals, and short- or medium-term funding was asked for by 8% of the journals. More staff resources were requested by 10% of the journals (Bosman et al. 2021). Bosman et al. also noted that:

"Some respondents underlined that it is an ethical responsibility for funders to support or even favour non-profit journals by providing them with some basic level of support" (Bosman et al. 2021, 120).

It should be noted here that in this study lots of types of journals reported to be under the banner of DOA. These have organisational practices ranging through small voluntary run journals, learned society journals, institutional journals, commercial and non-commercial publisher journals, and large professional journals (Bosman et al. 2021).

Personal time input into the day-to-day functioning of new online publishing houses and journals could be collective and collaborative, which would create an organisation based on shared responsibility. However, there is the question of where the bulk of the editorial staff time will come from, as academics working in higher education already have high workloads. This prompts the question as to whether higher education institutions would be willing to donate a percentage of staff time to use as editorial time. Research universities are involved in the generation of knowledge and are expected to operate in a sustainable and socially responsible way (Nejati et al. 2011). Their contribution would aid combating the predatory journal problem and would demonstrate a level of corporate social responsibility on their part. Alternatively, it might be possible that editors could donate a proportion of the editorial time from their own time, and the rest of their editorial time could be donated by their host higher education institution in the form of work planned university allocated time.

This idea comes with a substantial caveat. If universities donate staff time to use as editorial time, then it would ideally need to be made mandatory in some way, as a voluntary donation could easily be withdrawn. This would be very similar to a form of public funding. Thus, a more practical alternative would be for a specific funding body to be created that offers funding for a 5 to 10-year period to cover those costs of running a DOA journal, that cannot be covered by practices involving collaboration, mutual aid, and the social anarchist economies described above.

A means of nurturing the DOA model needs to be incorporated into large scale policy making decisions. Taxes are often used for state funding of universities and to facilitate the payment student fees. Universities and libraries spend some of this money on subscriptions to journals. Moreover, some funding institutions and/or research councils offer publishing subsidies that can also end up being paid to for-profit publishers. Thus, there is evidence that public funds end up either directly, or indirectly, funding for-profit publishers (Fuchs and Sandoval 2013). This practice needs to be stopped as it facilitates the capitalist exploitation of authors and readers. Indeed, Kropotkin (1906) envisaged a future where:

"Literature and journalism will cease to be a means of money-making and living at the cost of others" (Kropotkin 1906, 48).

Publishers of predatory journals also see research literature as a means of making money, and thus thrive at the expense of others. The present discussion has focused on using components of the anarchist communism proposed by Kropotkin (1902, 1906) as a small-scale social paradigm that can inform the development of a mutual aid-based intervention to the predatory journal problem. In summary, although much of the work required can be based upon the anarchist(ic) activity of mutual aid based collaborative networks, monetary funding will always be required for work relating to

the purchase and maintenance of electronic and online resources. Such funding could also help pay for additional editorial time.

At present, anarchist(ic) run academic journals must exist as pockets of resistance within a selection of broader capitalist type economies, although they would still exist within any possible future democratic socialist economy (such as that favoured by Sunkara 2020). Not all social anarchist collectives are likely to agree with relying on public funding. However, DOA publishing could be supported if public funding were available to subsidise editorial, proof-reading and technical staff (Fuchs and Sandoval 2013). Moreover, public funding for DOA publishing by autonomous journals might be more readily available in a possible future democratic socialist economy (such as that favoured by Sunkara 2020) than in any of the current capitalist type economies.

8. Conclusion

Academic publishing has become a capitalist endeavour (Luescher and van Schalkwyk 2018). It seems likely that this capitalist endeavour has encouraged the development of pay-to-publish journals, which has in turn proliferated the manifestation of predatory journals. Thus, the present paper has discussed some anti-capitalist political ideologies, including economic practices relating to social anarchism that may inform a possible intervention to the predatory journal problem. In particular, the present paper has discussed the anarchist communism advocated by Pyotr Alexeyevich Kropotkin with a particular emphasis on an organisational behaviour referred to as mutual aid (Kropotkin 1902, 1906). In contemporary society mutual aid strategies are employed by left-wing political movements, and minority groups fighting for survival (Kenworthy et al. 2023). Mutual aid projects are often manifested as responses to crises (Spade 2020). The present paper has discussed how non-commercial not-for-profit online open access publishing houses and journals could evolve that are organised by faculty in higher education institutions using a mutual aid strategy. This process could be facilitated by the availability of funding for editorial, proof-reading and technical staff. It is entirely possible that a gradual and sustained increase in (anti-capitalist) online open access journals would result in a gradual and sustained decrease in (capitalist) predatory journals.

References

- Adema, Janneke and Samuel A Moore. 2018. Collectivity and Collaboration: Imagining New Forms of Communality to Create Resilience in Scholar-Led Publishing. *Insight* 31 (3): 1-11. DOI: <u>https://doi.org/10.1629/uksg.399</u>
- Bosman, Jeroen, Jan Erik Frantsvåg, Bianca Kramer, Pierre-Carl Langlais and Vanessa Proudman. 2021. The OA Diamond Journals Study. Part 1: Findings. Accessed 16 February 2025. DOI: <u>https://doi.org/10.5281/zenodo.4558704</u>
- Boucas, Dimitris. 2020. Theory, Reality, and Possibilities for a Digital/Communicative Socialist Society. *tripleC: Communication, Capitalism & Critique* 18 (1): 48-66. DOI: <u>https://doi.org/10.31269/triplec.v18i1.1140</u>
- Callaghan, Chris W. and Denise R. Nicholson. 2020. Predatory Publishing and Predatory Journals: A Critical Review and Proposed Research Agenda for Higher Education. *Journal of Further and Higher Education* 44 (10): 1433-1449. DOI: <u>https://doi.org/10.1080/0309877X.2019.1695762</u>
- Chiarelli, Andrea, Susan Murray and Ivonne Lujano. 2023. The Costs of Running an Online Open Access Journal. Open Access Journals Toolkit. DOI:

https://www.oajournals-toolkit.org/running-a-journal/the-costs-of-running-an-onlineopen-access-journal

Comedia (1984). The Alternative Press: The Development of Underdevelopment. Media Culture and Society 6 (2): 95-102. DOI:

https://journals.sagepub.com/doi/10.1177/016344378400600202

- Cornell, Andrew. 2016. Unruly Equality: U.S. Anarchism in the 20th Century. Berkeley, CA: University of California Press.
- Cress, Phaedra E and David B Sarwer. 2019. Predatory Journals: An Ethical Crisis in Publishing, *Aesthetic Surgery Journal Open Forum* 1 (1): 1-3. DOI: <u>https://doi.org/10.1093/asjof/ojz001</u>
- Darwin, Charles. 1859. On the Origin of Species by Means of Natural Selection, Or The Preservation of Favoured Races in the Struggle for Life. London: John Murray
- de Geus, Marius. 2014. Peter Kropotkin's Anarchist Vision of Organization. *Ephemera: Theory & Politics in Organization* 14 (4): 853-871. DOI: <u>https://ephemerajournal.org/sites/default/files/pdfs/contribution/14-4degeus_0.pdf</u>
- Dadkhah, Mehdi and Giorgio Bianciardi. 2016. Ranking Predatory Journals: Solve the Problem Instead of Removing It! *Advanced Pharmaceutical Bulletin* 6 (1): 1-4. DOI: <u>https://doi.org/10.15171/apb.2016.001</u>
- Demir, Selcuk Besir. 2018. Predatory Journals: Who Publishes in Them and Why? Journal of Informetrics 12 (4): 1296-1311. DOI: <u>https://doi.org/10.1016/j.joi.2018.10.008</u>
- Ferretti, Federico. 2016. Organisation and Formal Activism: Insights from the Anarchist Tradition. *International Journal of Sociology and Social Policy* 36 (11/12): 726-740. DOI: <u>http://dx.doi.org/10.1108/IJSSP-11-2015-0127</u>
- Ferris, Lorraine E and Margaret A Winker. 2017. Ethical Issues in Publishing in Predatory Journals. *Biochemia Medica* 27 (2): 279-84. DOI: https://doi.org/10.11613/BM.2017.030
- Fuchs, Christian and Marisol Sandoval. 2013. The Diamond Model of Open Access Publishing: Why Policy Makers, Scholars, Universities, Libraries, Labour Unions and the Publishing World Need to Take Non-Commercial, Non-Profit Open Access Serious. *tripleC: Communication, Capitalism & Critique* 13 (2): 428-443. DOI: <u>https://doi.org/10.31269/triplec.v11i2.502</u>
- Fuchs, Christian. 2020a. Communicative Socialism/Digital Socialism. tripleC: Communication, Capitalism & Critique 18 (1): 1-31. DOI: <u>https://doi.org/10.31269/triplec.v18i1.1144</u>
- Fuchs, Christian. 2020b. The Utopian Internet, Computing, Communication, and
- Concrete Utopias: Reading William Morris, Peter Kropotkin, Ursula K. Le Guin, and P.M. in the Light of Digital Socialism. *tripleC: Communication, Capitalism & Critique* 18 (1): 146-186. DOI: https://doi.org/10.31269/triplec.v18i1.1143
- Fuchs, Christian. 2021. The Digital Commons and the Digital Public Sphere: How to Advance Digital Democracy Today. Westminster Papers in Communication and Culture 16 (1): 9-26. DOI: <u>https://doi.org/10.16997/wpcc.917</u>
- Gordon, Uri. 2012. Anarchist Economics in Practice. In *The Accumulation of Freedom. Writings on Anarchist Economics*, edited by Deric Shannon, Anthony J Nocella II, and John Asimakopoulos, 204-218. Chico, CA: AK Press.
- Gunderson, Ryan. 2021. Must Cognitive Sociology Heed Capitalism? Attention and Marginal Consciousness in Political-Economic Context. *Journal for the Theory of Social Behaviour* 51 (3): 306-326. DOI: <u>https://doi.org/10.1111/jtsb.12269</u>
- Honneth, Axel. (2018). The Idea of Socialism. Cambridge: Polity Press.

- Hong, Sung-Tae and Ho Shik Youn. 2020. Status of Editing and Publishing of Scholarly Journals by Academic Societies of Science and Technology in Korea. *Journal of Korean Medical Science* 35 (25): 1-10. DOI: https://doi.org/10.3346/jkms.2020.35.e208
- Hyland, Ken. 2023. Academic Publishing and the Attention Economy. *Journal of English for Academic Purposes* 64: 1-10. DOI: https://doi.org/10.1016/j.jeap.2023.101253
- Jessop, Bob. 2018. On Academic Capitalism. *Critical Policy Studies* 12 (1): 104-109. DOI: <u>https://doi.org/10.1080/19460171.2017.1403342</u>
- Kenny, Kate. 2021. Academic Activism. *Ephemera: Theory & Politics in Organization* 21 (4): 203-215. DOI:

https://ephemerajournal.org/sites/default/files/pdfs/contribution/21.4Kenny.pdf

Kenworthy, Nora, Emily Hops and Amy Hagopian. 2023. Mutual Aid Praxis Aligns Principles and Practice in Grassroots COVID-19 Responses Across the US. *Kennedy Institute of Ethics Journal* 33 (2): 115-144. DOI: <u>https://doi.org/10.1353%2Fken.2023.a904080</u>

Kinna, Ruth. 1995. Kropotkin's Theory of Mutual Aid in Historical Context. International Review of Social History 40 (2): 259-283. DOI: https://www.jstor.org/stable/44583751

- Kinna, Ruth. 2019. *The Government of No One: The Theory and Practice of Anarchism*. London: Penguin Random House.
- Knoche, Manfred. 2020. Science Communication and Open Access: The Critique of the Political Economy of Capitalist Academic Publishers as Ideology Critique. *tripleC: Communication, Capitalism & Critique* 18 (2): 508-534. DOI: https://doi.org/10.31269/triplec.v18i2.1183
- Krawczyk, Franciszek and Emanuel Kulczycki. 2021. How is Open Access Accused of Being Predatory? The Impact of Beall's Lists of Predatory Journals on Academic Publishing. *The Journal of Academic Librarianship* 47 (2): 1-11. DOI: https://doi.org/10.1016/j.acalib.2020.102271

Kropotkin, Pyotr A. 1902. *Mutual Aid: A Factor of Evolution*. Penguin Random House, UK.

- Kropotkin, Pyotr A. 1906. The Conquest of Bread. Amazon.
- Kurt, Serhat. (2018). Why Do Authors Publish in Predatory Journals? *Learned Publishing* 31 (2): 141-147. DOI: <u>http://dx.doi.org/10.1002/leap.1150</u>
- Late, Elina, Laura Korkeamaki, Janne Pölönen, and Sami Syrjämäki. 2020. The Role of Learned Societies in National Scholarly Publishing. *Learned Publishing* 33 (1): 5-13. DOI: <u>https://doi.org/10.1002/leap.1270</u>
- Loacker, Bernadette. 2021. Challenging Thought at Ephemera: Attempting to Think and Organize Differently. *Ephemera: Theory & Politics in Organization* 21 (4), 143-162. DOI: https://ephemerajournal.org/sites/default/files/2024-05/21.4Loacker.pdf
- Lockett, A. & Speicher, L. (2016). New University Presses in the UK: Accessing a Mission. *Learned Publishing* 29 (1), 320-329. DOI: https://doi.org/10.1002/leap.1049
- Luescher, Thierry M and Francois van Schalkwyk. 2018. African University Presses and the Institutional Logic of the Knowledge Commons. *Learned Publishing* 31 (S1): 288-298. DOI: <u>https://doi.org/10.1002/leap.1187</u>
- Malherbe, Nick. 2024. A Psychoanalytic Case for Anti-capitalism as an Organisational Form. *Theory, Culture & Society* 41 (6): 77-94. DOI: https://doi.org/10.1177/02632764231178648

- Marx, Karl. 1844. Economic and Philosophic Manuscripts of 1844. In *MECW Volume* 3, 229-346. London: Lawrence & Wishart.
- Marx, Karl and Frederick Engels. 1969 /1848. Manifesto of the Communist Party. In Marx/Engels Selected Works Volume One, 98-137. Moscow: Progress https://www.marxists.org/archive/marx/works/download/pdf/Manifesto.pdf
- Nejati, Mehran, Azadeh Shafaei, Yashar Salamzadeh, and Mohammadreza Daraei. 2011. Corporate Social Responsibility and Universities: A Study of Top 10 World Universities' Websites. *African Journal of Business Management* 5 (2): 440-447. DOI: <u>https://ssrn.com/abstract=1868688</u>
- Niles, Meredith T, Lesley A Schimanski, Eric C McKiernan, and Juan P. Alperin. 2020. Why We Publish Where We Do: Faculty Publishing Values and Their Relationship to Review, Promotion and Tenure Expectations. *PLoS ONE* 15 (3): 1-15: e0228914. DOI: <u>https://doi.org/10.1371/journal.pone.0228914</u>
- O'Hearn, Denis and Andrej Grubačić. 2016. Capitalism, Mutual Aid, and Material Life: Understanding Exilic Spaces. *Capital & Class* 40 (1): 147-165. DOI: <u>https://doi.org/10.1177/0309816816628562</u>
- Padovan, Dario. 1999. Social Morals and Ethics of Nature: From Peter Kropotkin to Murray Bookchin. Democracy & Nature, The International Journal of Inclusive Democracy 5 (3): 1-19. DOI: https://www.democracynature.org/vol5/padovan_morals.htm
- Pantel, Jelena H, Selva Varengo and Federico Venturini. 2023. Kropotkin and Social Ecology: Between Biology and Revolution. *Anarchist Studies* 31 (2): 73-95. DOI: http://doi.org/10.3898/AS.31.2.04
- Papadimitropoulos, Vangelis and Haris Malamidis. 2024. The Transformative Potential of Platform Cooperativism: The Case of CoopCycle. *tripleC: Communication, Capitalism & Critique* 22 (1): 1-24. DOI: <u>https://doi.org/10.31269/triplec.v22i1.1418</u>
- Pölönen, Janne, Sami Syrjämäki, Antti-Jussi Nygård, and Björn Hammarfelt. 2021.
 Who are the Users of National Open Access Journals? The Case of the Finnish Journal.fi Platform. *Learned Publishing* 34 (4): 585-592. DOI: https://doi.org/10.1002/leap.1405
- Rice, Danielle B, Becky Skidmore, and Kelly D Cobey. 2021. Dealing With Predatory Journal Articles Captured in Systematic Reviews. *Systematic Reviews* 10 (175): 1-4. DOI: <u>https://doi.org/10.1186/s13643-021-01733-2</u>
- Sandoval, Marisol. 2020. Entrepreneurial Activism? Platform Cooperativism Between Subversion and Co-optation. *Critical Sociology* 46 (6): 801-817. DOI: <u>https://doi.org/10.1177/0896920519870577</u>
- Scholz, T. (2016). Platform Cooperativism: Challenging the Corporate Sharing Economy. New York: Rosa Luxemburg Stiftung. <u>https://rosalux.nyc/wp-</u> <u>content/uploads/2020/11/RLS-NYC_platformcoop.pdf</u>
- Spade, Dean. 2020. Solidarity Not Charity: Mutual Aid for Mobilization and Survival. Social Text 38 (1): 131-151. DOI: <u>https://doi.org/10.1215/01642472-7971139</u>
- Springer, Simon, Myriam Houssay-Holzschuch, Claudia Villegas, and Levi Gahman. 2017. Say 'Yes!' to Peer Review: Open Access Publishing and the Need for Mutual Aid in Academia. *Fennia* 195 (2): 185-188. DOI: <u>https://hal.science/hal-01645153</u>
- Stoner, Aklexander M. 2020. Critical Reflections on America's Green New Deal: Capital, Labor, and the Dynamics of Contemporary Social Change. *Capitalism Nature Socialism* 32 (2): 77-94. DOI: https://doi.org/10.1080/10455752.2020.1775860

68

- Sunkara, Bhaskar. 2020. The Socialist Manifesto. The Case for rRadical Politics in an Era of Extreme Inequality. London: Verso.
- Teixeira da Silva, Jaime A, Mina Moradzadeh, Kwabena Osei Kuffour Adjei, and Christopher M Owusu-Ansah, Mulubrhan Balehegn, Eduardo I Faúndez, Manthan D Janodia and Aceil Al-Khatib. 2022. An Integrated Paradigm Shift to Deal With "predatory" publishing. *The Journal of Academic Librarianship* 48 (1): 1-15. DOI: <u>https://doi.org/10.1016/j.acalib.2021.102481</u>
- Törnberg, Anton. 2021. Prefigurative Politics and Social Change: A Typology Drawing on Transition Studies, *Distinktion: Journal of Social Theory* 22 (1): 83-107. DOI: <u>https://doi.org/10.1080/1600910X.2020.1856161</u>
- Tötösy de Zepetnek, Steven and Joshua Jia. 2014. Electronic Journals, Prestige, and the Economics of Academic Journal Publishing. *CLCWeb: Comparative Literature and Culture* 16 (1): 1-12. DOI: <u>http://docs.lib.purdue.edu/clcweb/vol16/iss1/12</u>
- Wallace, W. Angus. 2019. Publish and Be Damned. The Damage Being Created by Predatory Publishing. *The Bone and Joint Journal* 101-B (5): 500–501. DOI: https://doi.org/10.1302/0301-620X.101B5.BJJ-2018-1330.R1
- West, Stuart A, Ashleigh S Griffin and Andy Gardner. 2007. Evolutionary Explanations for Cooperation. *Current Biology* 17 (16): 661-672. DOI: <u>https://doi.org/10.1016/j.cub.2007.06.004</u>
- Wigger, Angela. 2014. A Critical Appraisal of What Could be an Anarchist Political Economy. *Ephemera: Theory & Politics in Organisation* 14 (4): 739-751. <u>https://re</u> DOI: <u>pository.ubn.ru.nl/bitstream/handle/2066/133334/133334.pdf</u>
- Williams, Dana M. 2018. Contemporary Anarchist and Anarchist Movements. Sociology Compass 12 (6) Article e12582: 1-17. DOI: <u>https://doi.org/10.1111/soc4.12582</u>
- Wright, Erik Olin. 2021. How to Be an Anti-Capitalist in the 21st century. London: Verso.
- Wright, Erik Olin. 2010. Envisioning Real Utopias. London: Verso
- Yates, Luke. 2021. Prefigurative Politics and Social Movement Strategy: The Roles of Prefiguration in the Reproduction, Mobilisation and Coordination of Movements. *Political Studies* 69 (4): 1033-1052. DOI: https://doi.org/10.1177/0032321720936046

About the Author

Andrew R. du Rocher is a senior lecturer in psychology in the Institute of Social Sciences at Sheffield Hallam University. His teaching is usually focused on statistics, research methods, individual differences, mental health and cognitive processes. His PhD focused on anxiety and the cognitive control of visual affect. He has published research on various topics including individual differences, mental health, cognition and emotion, pedagogy, and dreaming. He is interested in how anti-capitalist political ideologies can inform an intervention to improve the current global academic publishing paradigm.